Author's response to reviews

Title: Global Health Education: a cross-sectional study among German medical students to identify needs, deficits and potential benefits (Part 1 of 2: Mobility patterns & educational needs and demands)

Authors:

Kayvan Bozorgmehr (kayvan.bozorgmehr@googlemail.com)
Kirsten Schubert (ki.schubert@googlemail.com)
Johannes Menzel-Severing (menzel-severing@web.de)
Peter Tinnemann (Peter.Tinnemann@charite.de)

Version: 8 Date: 6 October 2010

Author's response to reviews:

Dear Editorial Team,

The corresponding author has cited the sister manuscript in the reference list as well as in the manuscript as 'in press'. The reference number in the main manuscript text is [20], the other references have been re-numbered accordingly.

Best wishes
Kayvan Bozorgmehr